Author's response to reviews

Title: Study of Mental Activity and Regular Training (SMART) in at risk individuals: A randomised double blind, sham controlled, longitudinal trial

Authors:

Nicola J Gates (nicola.gates@student.unsw.edu.au)
Michael Valenzuela (michaelv@unsw.edu.au)
Perminder S Sachdev (p.sachdev@unsw.edu.au)
Nalin A Singh (nalin.singh@sswahs.nsw.gov.au)
Bernhard T Baune (bernhard.baune@icu.edu.au)
Henry Brodaty (h.brodaty@unsw.edu.au)
Chao Suo (z3196279@unsw.edu.au)
Nidhi Jain (nidhi.jain@sydney.edu.au)
Guy C Wilson (guy.wilson@sydney.edu.au)
Yi Wang (yiwang0602@hotmail.com)
Michael K Baker (michael.baker@ecu.edu.au)
Dominique Williamson (dominique.williamson@sydney.edu.au)
Nasim Foroughi (n.foroughi@usyd.edu.au)
Maria A Fiatarone Singh (msingh@usyd.edu.au)

Version: 4 Date: 16 April 2011

Author's response to reviews:

Please find revised submission with reference 24 now cited in text and reference list updated accordingly.